Author's response to reviews

Title: The QICKD study protocol: a cluster randomised trial to compare quality improvement interventions to lower systolic BP in chronic kidney disease (CKD) in primary care [ISRCTN56023731]

Authors:

Simon de Lusignan (slusigna@sgul.ac.uk)
Hugh Gallagher (Hugh.Gallagher@epsom-sthelier.nhs.uk)
Tom Chan (tchan@sgul.ac.uk)
Nicki Thomas (N.M.Thomas@city.ac.uk)
Jeremy van Vlymen (jvanvlym@sgul.ac.uk)
Michael Nation (michaelnation@kidneyresearchuk.org)
Neerja Jain (NeerjaJain@kidneyresearchuk.org)
Aumran Tahir (mtahir@nhs.net)
Elizabeth du Bois (Elizabeth.Dubois@wpct.nhs.uk)
Iain Crinson (icrinson@sgul.ac.uk)
Nigel Hague (njhmg@hotmail.co.uk)
Fiona Reid (freid@sgul.ac.uk)
Kevin Harris (Kevin.Harris@uhl-tr.nhs.uk)

Version: 5 Date: 1 May 2009

Author's response to reviews:

Dear Editor,

Revised version attached. I have accepted the copy editors changes and made three other tiny changes. One being that the other authors prefer QICKD (with a capital at the end!)

Thanks for your help

Simon de Lusignan